Author’s response to reviews

Title: Assessment and model guided cancer screening promotion by village doctors in China: a randomized controlled trial protocol

Authors:

Rui Feng (853571123@qq.com)
Xingrong Shen (1264854421@qq.com)
Jing Chai (jingchai82@sina.com)
Penglai Chen (orangechenpl@aliyun.com)
Jing Cheng (jing_hp@sina.com)
Liang Han (1010066583@qq.com)
Ting Zhao (1252509750@qq.com)
Rui Sha (1214984721@qq.com)
Kaichun Li (295240260@qq.com)
Debin Wang (dbwang@vip.sina.com)

Version: 8
Date: 5 August 2015

Author’s response to reviews: see over
Dear editor and reviewers,

Thank you very much for your careful review of the aforementioned manuscript. We had revised the manuscript as suggested and enclosed please find our item by item responses to all issues raised.

With best regards,

Sincerely,

Debin Wang
School of Health Service Management
Anhui Medical University
China.

Responses to reviewer 1, Dr Johanna E. Maree

Re: There are still minor language issues that need attention. Should quasi-experimental not be quasi-experiential? Should the initials of the authors be included in the citations? I think a second round of language editing would be of benefit.

Author responses: We changed “quesi” into “quasi”, deleted initials of the authors in the citations and asked help from a third English teacher in editing the language.

Responses to reviewer 2, Dr Qinyan Gao

Dr Qinyan Gao did not propose additional suggestions.

Responses to Additional Editorial Request

Re: We recommend that you copyedit the paper to improve the style of written English.

Author response: Given that native English speakers are not available from our university. We have applied for additional funding, from the project funding agency, to cover copyediting service from native English speakers as suggested. However, we may only be able to do so for future manuscripts. For the present manuscript, we asked help from a third English teacher in copyediting the paper.